Author’s response to reviews

Title: Effect of a multi-ingredient based food supplement on sexual function in women with low sexual desire. Pilot study.

Authors:

Santiago Palacios (ipalacios@institutopalacios.com)
Eva Maria Soler (eva.maria.soler@institutopalacios.com)
MARIETA RAMIREZ (marieta.ramirez@institutopalacios.com)
MARIELLA LILUE (mariella.lilue@institutopalacios.com)
Danial Khorsandi (khorsandi.d@procarehealth.com)
FERNANDO LOSA (losa.fernando@gmail.com)

Version: 3 Date: 22 Feb 2019

Author’s response to reviews:

Dear Dr. Masakazu Terauchi,

Editor of the BMC Women's Health,

We would like to cordially thank you and the reviewer for the constructive comment regarding the English grammar on our submission (BMWH-D-18-00307R1) to the Journal of BMC Women's Health. We have now thoroughly revised not only sections of added text but the entire manuscript. Hereby, we would like to enclose the revised final version of the manuscript.

If you have any questions, please do not hesitate to contact me at ipalacios@institutopalacios.com.

Thank you very much and looking forward to hearing from you.

Santiago Palacios, MD, Ph.D.
Reviewer reports:

C Rusike (Reviewer 2): Authors have addressed comments raised in the last review adequately. However, in sections of added text, English grammar is not up to standard and needs to be revisited and corrected to the discretion and satisfaction of the Editor.

We have done a complete revision of the English grammar of the entire manuscript.